Author's response to reviews

Title: Antioxidant activities and phenolic contents of the methanol extracts of the stems of Acokanthera oppositifolia and Adenia gummifera.

Authors:

Adeolu A Adedapo (adedapo3a@yahoo.co.uk)
Florence O Jimoh (asjimoh02@yahoo.com)
Anthony J Afolayan (aafolayan@ufh.ac.za)
Patrick J Masika (pmasika@ufh.ac.za)

Version: 3 Date: 8 August 2008

Author's response to reviews:

COVERING LETTER

Reviewer 1: Cephas Tagumirwa T Musabayane

Abstract

1. The abstract has been reduced to 350 words. It was not to be more than 350 words.

2. Objectives of the study are not to be part of the abstract. It appears this reviewer does not have all the details with respect to the editorial style of this journal.

Introduction

1. The objective of this study has now been clearly stated.

Results

1. The figures are described in percentiles hence there is no need to state these in standard error.

Reviewer 2: Fehmi Odabasoglu

Abstract

1. The results section of the abstract has now been corrected in line with the reviewer’s suggestion.